

The World's Largest Open Access Agricultural & Applied Economics Digital Library

# This document is discoverable and free to researchers across the globe due to the work of AgEcon Search.

Help ensure our sustainability.

Give to AgEcon Search

AgEcon Search
<a href="http://ageconsearch.umn.edu">http://ageconsearch.umn.edu</a>
aesearch@umn.edu

Papers downloaded from **AgEcon Search** may be used for non-commercial purposes and personal study only. No other use, including posting to another Internet site, is permitted without permission from the copyright owner (not AgEcon Search), or as allowed under the provisions of Fair Use, U.S. Copyright Act, Title 17 U.S.C.



# IFPRI Discussion Paper 00767

May 2008

# **Introducing a Genetically Modified Banana in Uganda**

Social Benefits, Costs, and Consumer Perceptions

Enoch Kikulwe, Wageningen University
Justus Wesseler, Wageningen University
and
José Falck-Zepeda, International Food Policy Research Institute

**Environment and Production Technology Division** 

## INTERNATIONAL FOOD POLICY RESEARCH INSTITUTE

The International Food Policy Research Institute (IFPRI) was established in 1975. IFPRI is one of 15 agricultural research centers that receive principal funding from governments, private foundations, and international and regional organizations, most of which are members of the Consultative Group on International Agricultural Research (CGIAR).

## FINANCIAL CONTRIBUTORS AND PARTNERS

IFPRI's research, capacity strengthening, and communications work is made possible by its financial contributors and partners. IFPRI gratefully acknowledges the generous unrestricted funding from Australia, Canada, China, Finland, France, Germany, India, Ireland, Italy, Japan, Netherlands, Norway, Philippines, Sweden, Switzerland, United Kingdom, United States, and World Bank.

#### Published by

# INTERNATIONAL FOOD POLICY RESEARCH INSTITUTE

2033 K Street, NW Washington, DC 20006-1002 USA

Tel.: +1-202-862-5600 Fax: +1-202-467-4439 Email: ifpri@cgiar.org

www.ifpri.org

#### **Notices**

Copyright 2008 International Food Policy Research Institute. All rights reserved. Sections of this material may be reproduced for personal and not-for-profit use without the express written permission of but with acknowledgment to IFPRI. To reproduce the material contained herein for profit or commercial use requires express written permission. To obtain permission, contact the Communications Division at <a href="mailto:ifpri-copyright@cgiar.org">ifpri-copyright@cgiar.org</a>.

<sup>&</sup>lt;sup>1</sup> Effective January 2007, the Discussion Paper series within each division and the Director General's Office of IFPRI were merged into one IFPRI–wide Discussion Paper series. The new series begins with number 00689, reflecting the prior publication of 688 discussion papers within the dispersed series. The earlier series are available on IFPRI's website at <a href="https://www.ifpri.org/pubs/otherpubs.htm#dp">www.ifpri.org/pubs/otherpubs.htm#dp</a>.

<sup>&</sup>lt;sup>2</sup> IFPRI Discussion Papers contain preliminary material and research results. They have not been subject to formal external reviews managed by IFPRI's Publications Review Committee but have been reviewed by at least one internal and/or external reviewer. They are circulated in order to stimulate discussion and critical comment.

# **Contents**

Acknowledgments	V
Abstract	vi
Abbreviations and Acronyms	vii
1. Introduction	1
2. The Relevance of a GM Banana for Uganda	3
3. Biosafety Regulations in Uganda	5
4. Toward Considering the Socioeconomic Aspects of a GM Banana	7
5. DataSsources, Data, and Preliminary Results	9
6. Willingness to Pay for a GM Banana	15
7. Summary, Conclusions, and Recommendations	21
References	24

# List of Tables

1.	The two dimensions of an ex ante analysis of social benefits and costs of GM crops	7
2.	Social benefits and costs for GM banana considered	9
3.	Incremental gross margin of cultivating one hectare of GM matooke	10
4.	Hurdle rates, average annual SIRBs and MISTICs per hectare of GM banana, per household, and per banana-growing farm household at different risk-free rates of return $(r)$ and risk-adjusted rates of return $(\mu)$	13
5.	Attributes, their definitions, and levels for choice sets	19
6.	Example of a choice scenario with a benefit presented	19
7.	Example of a choice scenario with no benefit presented	20
	Liet of Eigures	
	List of Figures	
1.	GM banana adoption rate over time	11

#### **ACKNOWLEDGMENTS**

The authors would like to thank Svetlana Edmeades, Melinda Smale, and Ekin Birol for their reviews of previous drafts of this paper. We would especially like to acknowledge all the contributions and ideas proposed by Dr. Tushemereirwe, leader of the Banana Research Program at the National Agricultural Research Organization (NARO) of Uganda, as well as those of other staff at NARO.

This paper was made possible through the support of Wageningen University and the International Food Policy Research Institute (IFPRI). In addition, support for this paper and for Enoch Mutebi Kikulwe's doctoral work was provided through grants from the Director General Office (DGO) Small Grants Initiative 2005 and funding from the European Commission for activities conducted by the Genetic Resources Policies Project (GRP-1) at IFPRI.

The authors would also like to acknowledge the support provided for this paper, field activities, and ongoing data collection and analysis on banana consumer and producer preferences in Uganda by the Program for Biosafety Systems (PBS) through the PBS Uganda Associate Award granted by the U.S. Agency for International Development (USAID) Uganda Mission and the award granted by the Office of the Administrator, Bureau for Economic Growth, Agriculture, and Trade/Environment and Science Policy, USAID, under the terms of Award EEMA-00-03-00001-00. The opinions expressed herein are those of the authors and do not necessarily reflect the views of USAID or its missions. This paper partially fulfills the requirements for Enoch Mutebi Kikulwe's doctor of philosophy degree at Wageningen University.

#### ABSTRACT

Banana is a staple crop consumed by Ugandan households. The Uganda National Agricultural Research Organization has implemented conventional and biotechnology programs that seek improving bananas and address the crop's most important pest and disease problems. A major thrust is the development of genetically modified (GM) bananas.

The purpose of this paper is to examine potential social welfare impacts of adopting a GM banana in Uganda. The study has three objectives. First, suggest and apply an approach to calculate reversible and irreversible benefits and costs of introducing a GM banana. The study applies a real option approach to estimate, ex ante, the maximum incremental social tolerable irreversible costs (MISTICs) that would justify immediate introduction of the technology. Second, suggest an approach for assessing producer/consumer preferences and willingness to pay (WTP) for introducing a GM banana. Finally, the paper discusses main implications for biosafety decision making for GM crops in Uganda.

Results of MISTICs estimation for different scenarios indicate that in delaying the approval of a GM banana, Uganda foregoes potential annual benefits ranging approximately from US\$179 million to US\$365 million. Average annual MISTICs per household vary between US\$34 and US\$ 69. Results indicate that only if the average household is willing to give up at least US\$38 per year to avoid introduction of a GM banana, should postponing an immediate release be considered. Results imply that although GM bananas promise vast benefits, realization of those benefits depends on consumers' perceptions and attitudes and the willingness to pay for the GM technology.

Keywords: GM banana, real option, choice experiment, biosafety, MISTICs, Uganda.

# **ABBREVIATIONS AND ACRONYMS**

IBC Institutional Biosafety Committee

MISTIC Maximum Incremental Social tolerable Irreversible Costs

NARO National Agricultural Research Organization
NBAC National Biotechnology Advisory Committee

NBC National Biosafety Committee

SIRB Social Incremental Reversible Benefit

UNCST Uganda National Council of Science and Technology

#### 1. INTRODUCTION

Banana is a staple crop in Uganda. Ugandans have the highest per capita consumption of cooking banana in the world (Clarke, 2003). However, banana production in Uganda is limited by several productivity constraints such as pests, diseases, soil depletion, and poor agronomic practices. To address those constraints, the country has invested significant resources in research and development and other publicly funded programs, pursuing approaches over both the short and long term. Uganda formally initiated its short-term approach in the early 1990s; it involves the collection of both local and foreign germ plasms for the evaluation and selection of cultivars tolerant to the productivity constraints. The long-term approach, launched in 1995, includes breeding for resistance to the productivity constraints using conventional breeding methods and genetic engineering. Genetic engineering projects in Uganda target the most popular and infertile cultivars that cannot be improved through conventional (cross) breeding. The main objective of genetic engineering in Uganda is to develop genetically modified (GM) cultivars that are resistant to local pests and diseases, have improved agronomic attributes, and are acceptable to consumers (Kikulwe et al., 2007).

The introduction of a GM banana in Uganda is not without controversy. In Uganda, where the technology of genetic engineering is still in its infancy, it is likely to generate a wide portfolio of concerns, as it has in other African countries. According to the Uganda National Council of Science and Technology (UNCST) (2006), the main public concern is the safety of the technology for the environment and human health.

Several countries have designed and implemented policies to address the safety concerns of consumers and producers (Beckmann, Soregaroli, and Wesseler, 2006a, 2006b). Such policies include assessment, management, and communication of the biosafety profiles of genetically modified organisms (GMOs) (Falck-Zepeda, 2006). As a consequence of its international obligations and the need to guarantee a socially accepted level of safety to its citizens, Uganda has taken significant steps to ensure the safety of GM biotechnology applications. GM banana varieties will need to undergo biosafety assessments<sup>1</sup> and receive the regulatory approval of the country's National Biosafety Committee before being approved for research, confined field trials, and release into the environment for commercialization.<sup>2</sup>

The biosafety regulatory process, however, has several economic consequences as biosafety regulations are not costless endeavors. Kalaitzandonakes, Alston, and Bradford (2007) calculate the compliance costs for regulatory approval of herbicide-tolerant and insect-resistant maize to be on the order of about US\$7 to 50 million. They note that the approval costs for similar types of GM crops will be similar. In addition, biosafety-testing requirements can consume significant amounts of time—from a few months to several years. A delay in the approval of a new variety forestalls access to the potential benefits generated by farmer adoption of the technology, and one can expect such costs to be substantially higher than the regulatory compliance costs. Wesseler, Scatasta, and Nillesen (2007), for example, estimated that the average annual benefits of Bt corn for the European Union amount to about 155 million euros per year. On the other hand, regulatory processes create additional information about the technology and can help to improve the selection and regulation of appropriate technologies.

Jaffe (2006) has noted that existing drafts of Uganda's biotechnology and biosafety policy stress the importance of the socioeconomic implications of the technology for biosafety regulation, but that author also observes a lack of precision in identifying the socioeconomic aspects and how they should be

<sup>&</sup>lt;sup>1</sup> The original scope of biosafety as described in the Cartagena Protocol on Biosafety was environmental safety. However, over time the original scope has been expanded to include food and feed safety in terms of toxicity or allergenicity. In this paper, it is therefore understood that the label biosafety includes both environmental and food/feed safety.

<sup>2</sup> As of the final draft of this paper, three technologies have been approved for confined field trial testing in Uganda, including a virus-resistant cassava, an insect-resistant and herbicide-tolerant cotton, and a fungus-resistant banana. The cotton and banana confined field trials are being proposed by NARO. Unlike the confined cotton field trial, the banana confined field trial including 200 plantlets has been established on 30th November 2007.

considered. Each country decides independently whether to include socioeconomic considerations as part of the process of deciding which technology may be approved for commercialization after being deemed safe by the biosafety authority. In fact, Article 26.1 of the Cartagena Protocol gives countries the choice of whether to include socioeconomic considerations in the biosafety assessment process consistent with other international treaties although limited to the context of biodiversity (Jaffe, 2006). Article 26.1's "may take into account" clause has been applied strictly in some countries, such as India, where the socioeconomic consideration is mandatory for biosafety applications.

Many countries, including Uganda, have not determined whether and how to include socioeconomic considerations, at what stage of the regulatory process to include them, and what the scope and decision-making process within biosafety regulations should be. In fact, some biosafety experts (and some countries) have resisted including such considerations in the biosafety decision-making process, as in their view, such issues may cloud that process and distract regulators from the scientific/technical issues related directly to biosafety. It is worthwhile to note that inclusion of socioeconomic considerations for biosafety regulatory approval at the laboratory/greenhouse or confined field trial stages contributes very little to the decision-making process, as the material will not enter the food chain and thus will not be commercialized until it is given regulatory approval further along in the process. Therefore, a major objective of this discussion paper is to illustrate the relevance of socioeconomic analyses for supporting biotechnology decision making but also for contributing to the development and implementation of biosafety regulations. We present a general approach using GM banana as an example.

In the following sections we discuss the benefits that a GM banana could provide to producers and consumers in Uganda and the role of biosafety regulations in governing the introduction of a GM banana. A real option model is presented that shows how concerns about environmental risks can be considered within a cost-benefit analysis as a first step toward a socioeconomic assessment of introducing a GM banana in Uganda. We explicitly show how the underlying trade-offs between potential irreversible and reversible benefits and costs that accompany a GM banana can be assessed, building on previous research on banana production in Uganda by Bagamba (2007) and Edmeades and Smale (2006). We calculate the maximum incremental social tolerable irreversible costs (MISTICs). The application of the MISTICs approach pays closer attention to the application of the precautionary principle within the assessment of GM crops (Just, Alston, and Zilberman, 2006). It is important to note here that this is the first application of the MISTICs approach in a developing-country setting.

In addition, we show how the results of the economic analysis can be combined with the consumers' willingness to pay (WTP) for a GM banana using a choice experiment model. We explicitly demonstrate how one can use a latent segment model to capture and account for heterogeneity among consumer preferences given a tangible economic benefit of the GM banana.<sup>3</sup> The latent segment model complements and extends the dimensions of previous research (Li et al., 2003; Loureiro and Bugbee, 2005; Knight et al., 2007) on consumers' WTP for GM food by, first, incorporating the foregone economic benefits of a delay in release and, second, incorporating producers as consumers in the sample. The approach is unique in its application to banana varieties in a developing-country context.

The paper is structured as follows. The second section discusses in more detail the relevance of a GM banana for Uganda. Section 3 introduces an overview of biosafety regulations in Uganda. Section 4 presents the MISTICs approach and explains its application. Section 5 reports and discusses the preliminary results. Section 6 introduces the theoretical framework of the choice experiment and its application. The final section draws conclusions and discusses implications for decision making regarding biotechnology and biosafety regulations for a GM banana in Uganda.

2

<sup>&</sup>lt;sup>3</sup> A tangible economic benefit refers to a benefit foregone if the GM banana is not introduced immediately.

## 2. THE RELEVANCE OF A GM BANANA FOR UGANDA

Banana is one of the most important crops in Uganda with approximately 7 million people, or 26% of the population, depending on the plant as a source of food and income. Bananas are estimated to occupy 1.5 million hectares of the total arable land, or 38% of the cultivated land, in the country (Rubaihayo and Gold, 1993; Rubaihayo, 1991). The plant is grown primarily as a subsistence crop in rural areas, although consumption is not limited to rural areas as approximately 65% of urban consumers in Uganda have a meal of the cooking variety of banana at least once a day. Ugandans have the highest per capita consumption of cooking banana in the world (Clarke, 2003).

Most of the banana varieties grown in Uganda are endemic to the East African highlands—a region recognized as a secondary center of banana diversity (Stover and Simmonds, 1987; Swennen and Vuylsteke, 1988; Smale and Tushemereirwe, 2007). The endemic banana varieties (AAA–EA genomic group) consist of two use-determined types: cooking bananas (matooke) and beer bananas (mbidde). Karamura (1998) recognized 238 names of East African highland banana varieties in Uganda, with 84 clones grouped into five clone sets. The nonendemic clones include dessert bananas (varieties that are consumed raw), some beer bananas (varieties suitable for beer and juice making), and roasting bananas (or plantains).

Banana yields in Uganda are severely reduced by several pests and diseases. Among the pests that cause the most yield damage are weevils (*Cosmopolites sordidus*) and nematodes (*Radopholus similis*, *Pratylenchus goodeyi, and Helicotylenchus multicinctus*). The diseases that contribute to the worst yield losses in Uganda are the soil-borne fungal Panama disease, or *Fusarium* wilt (*Fusarium oxysporum*), bacterial wilts including the banana *Xanthomonus* wilt (*Xanthomonus campestris* pv. *musacearum*), and the air-borne fungal black leaf spot disease or "black Sigatoka" (*Mycosphaerella fijiensis* Morelet) (Gold, 1998; 2000; Gold et al., 1998, ; 2001; Tushemereirwe et al., 2003).

Consequently, the National Banana Research Program of the National Agricultural Research Organization (NARO) in Uganda has developed a breeding program that employs a range of traditional crop breeding methods and a portfolio of biotechnologies to address the crop's most debilitating problems caused by pests and diseases (Kikulwe et al., 2007). The short-term breeding strategy includes the assembly of local and foreign germ plasms for evaluation and selection of varieties resistant or tolerant to existing productivity constraints. Resistance to a limited set of pests and diseases (e.g., black Sigatoka) was identified in hybrid banana varieties. Though characterized by bigger bunches, the hybrid varieties are not widely grown in Uganda (Nowakunda, 2001; Smale and Tushemereirwe, 2007). Producers and consumers prefer the East African highland cooking bananas, but these are also highly susceptible to black Sigatoka (Nowakunda et al., 2000; Nowakunda, 2001) and bacterial wilts (Tushemereirwe et al., 2003). Susceptibility to diseases prompted the national researchers to adopt a long-term breeding strategy that includes the generation of new genotypes and other new approaches to introduce resistance.

The highest-yielding highland cooking bananas proved to be sterile, which slows down their improvement through conventional breeding (Ssebuliba, 2001; Ssebuliba et al., 2006). With major biotic constraints not easily addressed through conventional breeding and management practices, recent efforts have been made to employ genetic engineering for the insertion of resistance traits into selected banana background planting material. Unlike crossbreeding, genetic engineering allows for improving the agronomic traits (e.g., disease and pest resistance) as genes are inserted into potential host varieties (cultivars) while not changing other production and product attributes (e.g., cooking quality). The genetic modification approach has shown potential for the improvement of the crop (Tripathi, 2003). At the University of Leuven, Belgium, GM bananas with resistance against black Sigatoka have been developed. Yet the performance of the new varieties and/or traits inserted into local host varieties cultivated under local conditions is not known as the field trials have just begun.

Edmeades and Smale (2006) argue that the choice of a host variety for a genetic transformation largely determines its acceptability by producers and consumers. In those regions strongly affected by biotic constraints, it is likely that GM banana cultivars will be more beneficial to poorer and subsistence-

oriented farmers. In addition, the insertion of multiple traits into East African highland bananas, although associated with additional research and development costs (e.g., transformation costs, regulatory costs), could further increase the benefits generated by the adoption of the technology in Uganda. Multiple traits may also increase adoption rates, as farmers may not immediately notice the beneficial effect of a single trait.

Although GM bananas look promising for large-scale (mass clonal) multiplication and dissemination, empirical evidence of the success of such organisms is still limited. Long-term multiplication of micropropagated (tissue-cultured) plants, for example, may lead to epigenetic<sup>4</sup> (somaclonal) variations. Additionally, genetic uniformity in a trait intensifies the probability of mutations in the targeted pest or disease that overcome resistance and increase epidemic vulnerability. These two aspects raise questions about the clonal fidelity of offspring plants and their genetic stability, both affecting economic benefits of GM banana varieties. In this context biosafety measures to monitor, evaluate, and mitigate effects of such occurrences become critical for the appropriate deployment of the technology in Uganda.

<sup>&</sup>lt;sup>4</sup> Epigenetic changes are changes that do not affect the DNA sequence of genes but change the gene in other ways. These changes may be induced spontaneously in response to environmental factors or in response to the presence of a particular allele, even if it is absent from subsequent generations. Modgil et al. (2005) note that the in vitro process, length of in vitro culture, and in vitro stress due to unnatural and nutritional conditions are some of the factors believed to induce epigenetic changes.

# 3. BIOSAFETY REGULATIONS IN UGANDA

Uganda is among the few African countries that have invested in agricultural GM crop research and development and have initiated procedures for confined field trials to evaluate GM technologies (Atanassov et al., 2004).<sup>5</sup> The country has taken significant steps to ensure safety in biotechnology application (Nampala, Mugoya, and Sengooba, 2005). Biosafety regulations and, to a degree, biotechnology developments in Uganda are governed in the context of the Cartagena Protocol on Biosafety (GOU, 2002b; 2004). The country signed the protocol in May 2000 and ratified it in November 2001 (GOU, 2004; Wafula and Clark, 2005). UNCST is the institution responsible for the implementation of the biosafety protocol and the protocol's designated competent authority. UNCST established the National Biosafety Committee (NBC), a technical evaluation arm, in 1996. NBC is responsible for reviewing applications and implementing general biosafety guidelines and regulations (GOU, 2004; Wafula and Clark, 2005).

Currently, the basis for the development and application of biotechnology is the National Science and Technology Policy of 2001 (GOU, 2004). The National Science and Technology Policy provides general reference to biotechnology within the broader context of the role of science and technology in national development. The responsibilities of the various institutions and agencies involved in the approval process for biotechnology products have been outlined in the Biosafety Framework of 2000. The National Biosafety Framework, which was developed by UNCST, is based on the United Nations Environment Programme's International Technical Guidelines on Safety in Biotechnology (2001). Those guidelines provide terms of reference for NBC and the institutional biosafety committees (IBCs)—detailed information on risk assessment and management procedures for microbes, inspection, and approval (Traynor, 2003).

Under the National Biosafety Framework, UNCST has the mandate of approving genetically modified organisms for research purposes, confined release into the environment, and commercial planting in Uganda. UNCST receives all applications for research on or the deliberate introduction of GMOs, conducts a screening for completeness, and enters the applications into the national public records as submitted, before forwarding them to NBC for review and risk assessment evaluation. The complete risk assessment is done by the applicant. NBC is obliged to review the risk assessment dossiers submitted by the applicant after the application has been assessed by the IBCs and finally advises UNCST.

The members of NBC are stakeholders, such as representatives from regulatory agencies, the scientific community, universities, the private sector, and civil society. (Nampala, Mugoya, and Sengooba, 2005). NBC is also responsible for writing the draft National Biotechnology and Biosafety Policy, the draft National Biosafety Regulations (GOU, 2004), and the Guidelines on Biosafety in Biotechnology for Uganda (GOU, 2002a), and for developing draft manuals addressing specific issues surrounding biosafety regulations, such as confidential business information. These documents make up the biosafety regulatory framework for Uganda. At this point in time, the documents still need to be approved by the government (Jaffe, 2006). The implementation of the finally agreed upon biosafety regulations for a specific GM crop will be managed by UNCST. UNCST is advised by the National Biotechnology Advisory Committee (NBAC), which is an interministerial committee including representatives from key institutions in biotechnology development and NBC (GOU, 2004).

Applications for the import and export of GMOs are also approved by NBC. The government of Uganda (2005, p. 6) stipulates that "any person, prior to intentionally introducing a GMO shall apply to the Competent Authority for authorization. In case of imports, the exporter or the Competent Authority in the country of export may submit an application on behalf of an applicant, and may designate in the application with whom the Competent Authority shall communicate regarding the application." In the

5

<sup>&</sup>lt;sup>5</sup> The GM banana field trials were approved by the National Biosafety Committee and have been established at Kawanda by NARO. Note that Uganda is now among the other five African countries that have conducted confined field trials of GM crops namely: South Africa, Egypt, Burkina Faso, Zimbabwe, and Kenya. Of those, only South Africa has approved crops for commercialization.

case of field testing, the Ugandan government (2005, p. 7) continues to specify that "the applicant shall document to the Competent Authority that participating personnel will have appropriate training that the field test will be overseen by an individual possessing appropriate technical expertise." The competent authority has to reply within 90 days. Within a period of not more than 270 days after the scientific risk assessment, the competent authority will make a final decision on whether to approve or deny the applicant the authority to introduce the GMO. This implies that if all the required documents are submitted and complete, the approval will take about one year. In the case of a denial of a request, the applicant is given 30 days to appeal with genuine reasons and/or additional relevant information, and the competent authority has 30 days within which to render a final decision. The competent authority will finally avail to the public any proposal about the intentional introduction of GMOs. The public is given not more than 60 days to submit comments, which the competent authority shall take into consideration before the final decision is made.

According to Wafula and Clark (2005), NARO submitted applications to UNCST in 2000 to introduce Bt cotton and Bt maize, but they were not approved for confined field trials. One of the reasons UNCST gave was that Uganda was unprepared to handle GM crops because it lacked a national biotechnology and biosafety policy. In addition, Uganda lacked confinement and containment facilities for GMO field trials. Recently, the government selected biotechnology as one of the priority areas in its plan for the modernization of agriculture (Oxford Policy Management (OPM), 2005). Consequently, substantial investments have been made in institutional development and capacity building for agricultural biotechnology and biosafety.

Jaffe (2006) analyzed and compared biosafety regulatory systems in Uganda, Kenya, and Tanzania using the African Model Law, which is one of the documents given consideration throughout Africa when a country begins drafting laws and regulations to address biosafety. Jaffe's assessment considered key issues such as comprehensiveness, transparency, participation, and efficiency of the regulatory systems. In the context of Uganda's existing biosafety regulatory framework, despite all the efforts that have been made, the author notes some shortcomings, particularly in the areas of transparency and clarity regarding the process to be followed such that all interested stakeholders are able to understand and meet the requirements of the regulatory process. Those shortcomings include, first, no clear indications of how the assessment of the potential food safety risks that might arise from the GMOs will be handled. Second, even though UNCST is involved in the formulation of the biosafety regulation policy, the statute authorizing its creation does not provide legal authority to regulate GMOs. Neither the biosafety policy nor the government regulations establish a clear safety standard for approving a GMO. Third, within the documents, there are no elaborations on how and what socioeconomic considerations will be considered, how they will be analyzed, by whom, and how they will be considered in the decisionmaking process. These shortcomings will need to be addressed to ensure a robust, efficient, cost-effective, and socially acceptable biosafety regulatory process in Uganda.

## 4. TOWARD CONSIDERING THE SOCIOECONOMIC ASPECTS OF A GM BANANA

The economic net benefits of introducing a GM banana depend on the reversible and irreversible benefits and costs the technology will generate. Reversible benefits and costs can be defined as those benefits and costs that can be reversed after the planting of the crop and do not result in additional ex post (after stopping production) benefits and costs. An illustrative example is the purchase of inorganic fertilizer. If the producer finds that producing a GM banana crop is no longer worthwhile, the purchased fertilizer can be used for other crops. Similarly, other variable costs can be considered as being reversible as well.

On the other hand, irreversible benefits and costs refer to those benefits and costs that will continue to occur even if GM bananas are no longer produced or those that cannot be fully reversed. Examples are sunk costs or chronic health damages from pesticide use. The reversible and irreversible benefits and costs can be further differentiated into private and nonprivate benefits and costs. This differentiation is useful for understanding the distribution of benefits and costs between, for example, farmers (private) and society at large (nonprivate). The private and nonprivate costs include effects on nontarget species; for instance, the introduced genes in nematode-resistant GM banana may affect beneficial nontarget nematodes. Others include effects on human health such as antibiotic resistance and allergies, evolution of pests and disease resistant to the inserted genes (Kendall et al., 1995), and loss of genetic diversity (FAO, 2001). Certainly, a net reduction in the use of insecticides and nematicides on GM banana will have positive impacts on human health, the environment, and biodiversity, and those can be considered as being irreversible benefits (Wesseler, 2003). Demont, Wesseler, and Tollens (2004) provide a number of examples illustrating the difference between reversibility and irreversibility.

The different types of benefits and costs are summarized in Table 1. Table 1 shows a two-dimensional matrix differentiating between reversible and irreversible and private and nonprivate benefits and costs for an ex ante economic analysis of GM crops. The sum of quadrants one and two gives the value of the net social reversible benefits and that of quadrants three and four the net social irreversible costs. The irreversible costs are of critical importance for biosafety decision making. They are the major argument supporting biosafety regulations under the Cartagena Protocol on Biosafety (Secretariat of the Convention on Biological Diversity, 2000). However, it is not irreversibility itself that has been used exclusively to justify specific biosafety regulations for GM crops as well as to justify a delay in release (flexibility) to obtain additional knowledge and information on the new technology; rather, uncertainty about irreversible costs in combination with uncertainty about the economic benefits of GM crops has been put forward in the Cartagena Protocol and other regulatory processes to justify such interventions.

Table 1. The two dimensions of an ex ante analysis of social benefits and costs of GM crops

Reversibility	Scope	Private	Nonprivate
		Quadrant 1	Quadrant 2
Reversible		Private reversible benefits	External reversible benefits
		Private reversible costs	External reversible costs
		Quadrant 3	Quadrant 4
Irreversible		Private irreversible benefits	External irreversible benefits
		Private irreversible costs	External irreversible costs

Source: Demont, Wesseler, and Tollens (2004).

In the context of the Cartagena Protocol, the introduction of a new GM crop becomes a decision-making process under uncertainty, irreversibility, and flexibility. Analyzing decision making under uncertainty, irreversibility, and flexibility is not new to economists and has a tradition in environmental economics that originated in the early 1970s with papers published by Arrow and Fisher (1974) and

Henry (1974), while in economics it can even be traced back to Louis Bachelier (1900) (Bernstein, 1992). Irreversible benefits and costs in combination with uncertainty and flexibility can be considered within a real option approach for the assessment of the adoption impacts of a GM crop. Examples are provided by Demont, Wesseler, and Tollens (2004) and Scatasta, Wesseler, and Demont (2006) for the introduction of GM strains of sugar beet and corn in the European Union.

#### Maximum Incremental Social Tolerable Irreversible Costs

We begin with the assumption that incremental reversible net benefits follow a continuous-time, continuous-state process with trend, where GM crops may be released at a point in time. In this approach, the social incremental reversible benefits  $W^*$  (the sign \* indicates optimal timing) need to be greater than the difference between the social incremental irreversible costs (I) and the social incremental irreversible benefits (R), weighted by the size of the uncertainty and flexibility (or hurdle rate) associated with the

introduction of the new technology. The hurdle rate is commonly expressed in the form  $\frac{\beta}{\beta-1}$ , where

 $\beta > 1$  captures the uncertainty and flexibility effect and is a result of identifying the profit-maximizing decision rule under irreversibility, uncertainty, and flexibility, if benefits do follow a geometric Brownian motion.<sup>6</sup> The interpretation of the decision rule for the case of the GM banana is that, as long as

 $W - \frac{\beta}{\beta - 1}(I - R) \le 0$ , Uganda should delay adoption of a GM banana until more information about the new technology is available.

In the context of GM crops, where people are more concerned about the not-so-well-known irreversible costs of the technology, it is feasible to estimate threshold values that indicate the maximum incremental social irreversible costs that an individual or society in general is willing to tolerate as compensation for the benefits of the technology. Scatasta, Wesseler, and Demont (2006) have called this threshold value the *maximum incremental social tolerable irreversible costs*, *I\**, or *MISTICs* for short. In the specific case of Uganda, the estimated MISTICs can be interpreted as the maximum willingness to pay (WTP) for having the GM banana approved for planting in the country. Actual incremental irreversible social costs, *I*, are to be no greater than the sum of incremental irreversible social benefits and incremental reversible social net benefits for introducing a GM banana, such that

$$I < I^* = \frac{W}{\beta/(\beta - 1)} + R. \tag{1}$$

Using Equation 1 with parameter values generated for the case of GM banana can provide threshold values for the irreversible costs. The values can be compared with information from secondary sources to identify whether the threshold value will be met in Uganda.

In practice, estimation of the maximum incremental social tolerable irreversible costs (MISTICs, or  $I^*$ ) requires quantification of three factors:

social incremental reversible benefits from GM crops (SIRBs, or W); social incremental irreversible benefits (SIIBs, or R) rate; and hurdle rate,  $\beta/(\beta-1)$ .

All these factors can be estimated or calculated using econometric and mathematical modeling techniques following Demont, Wesseler, and Tollens (2004).

<sup>6</sup> The geometric Brownian motion is a Wiener process with a geometric trend for which changes expressed as natural logarithms are normally distributed. The Wiener process is a continuous-time, continuous-state stochastic Markov process with three properties: (a) probability distributions of future values depend on the current value only; (b) the Wiener process grows at independent increments; and (c) changes are normally distributed. The assumption that the adoption of this technology follows a geometric Brownian motion accounts for the uncertainty of the technology (Cox and Miller, 1965).

# 5. DATA SOURCES, DATA, AND PRELIMINARY RESULTS

Secondary data have been used for the estimations of parameters for the real option model. Data are taken from the database of a NARO/IFPRI project conducted between 2003 and 2004 in Uganda. The data set is complemented by data from the Uganda Bureau of Statistics (UBOS) and the Food and Agriculture Organization. Table 2 lists the private and nonprivate reversible and irreversible benefits and costs directly and indirectly considered.

Table 2. Social benefits and costs for GM banana considered

	Scope	
	Private	Nonprivate
Reversibility		
	Quadrant 1	Quadrant 2
	<u>Benefits</u>	<u>Benefits</u>
	- Higher yields	- Zero
Reversible		
	<u>Costs</u>	<u>Costs</u>
	- Labor costs	- Zero
	Quadrant 3	Quadrant 4
	<u>Benefits</u>	Benefits
	- Negligible	- Indirect: improved food safety and
		decreased vulnerability
Irreversible	<u>Costs</u>	
	- Planting material	<u>Costs</u>
	-	- Indirect: possible health and
		environmental effects

Source: Authors

The social incremental reversible benefits (SIRBs) were estimated based on private net benefits. Private incremental reversible benefits can be defined as the difference between the gross margin from GM and non-GM bananas, excluding planting material. Table 3 shows preliminary results for the incremental benefits estimations for a GM banana in Uganda. The starting point for these estimations is the gross margin for a non-GM banana crop as reported by Bagamba (2007, p. 31). The annual variable costs for a non-GM banana crop include hired labor used mainly for weeding and crop sanitation. The use of other inputs such as fertilizer and pesticides is, according to Bagamba, negligible. The average output in metric tons per year is about 10.6 per hectare with an average price about 149,600 Uganda shillings (USh) per metric ton. Under the current production practices, most farmers do not incur costs for planting materials. Most of the planting materials are exchanged for free between farmers (Kikulwe et al., 2007).

The average annual gross margin from producing one hectare of non-GM banana (traditional) is approximately USh 1,411,200 (US\$800) excluding labor costs for planting. The main benefit of introducing a GM banana is an increase in banana yield through reduced biotic pressure. Assuming that planting a GM banana with a gene resistant to black Sigatoka<sup>7</sup> increases yield by 20% and labor costs by about 10% and that the average annual costs for planting material are USh 151,700, the gross margin per

<sup>7</sup> In Uganda and in other countries, black Sigatoka may damage production, reducing yields up to 50% (Tushemereirwe et al., 2000; Craenen, 1998).

hectare would increase from about USh 1,411,200 to about USh 1,650,500, or by about USh 239,300. If the irreversible planting costs are deducted, the expected average private incremental reversible benefits are about USh 389,200 per hectare (about US\$222 per hectare).

Table 3. Incremental gross margin of cultivating one hectare of GM matooke

Variable	Non-GM banana (matooke)*	GM banana (matooke)**
Output (metric tons/year)	10.6	12.7
Price per ton (K)	149.6	149.6
Value of output (K)	1,504.4	1,902.9
Hired labor (hours)	232.8	256.1
Family labor (hours)	2,295.8	2,525.4
Total labor (hours)	2,528.6	2,781.5
Wage rate (USh/hour)	400.1	400.1
Cost of hired labor (K)	93.2	102.5
Cost of planting materials (K) <sup>a</sup>	0	151.7
Gross margin (K)	1,411.2	1,648.7
Return to family labor (USh/hour)	614.7	652.9
Expected average incremental gross margin (K)		237.5
Expected average private incremental reversible benefits (K)		389.2
Incremental average return per family labor (USh/hour)		38.2
Total incremental labor income per hectare (K)		96.4

<sup>\*</sup> Source: Bagamba (2007).

Note: Benefits and costs are valued in Uganda shillings (USh  $1,750 \approx \text{US}\$1$ ); return to fixed resources (e.g., land) is not deducted from the gross margin in the computation of return to family labor.

The introduction of a GM banana will trigger an additional cost for planting materials. The total planting costs for about 1,100 plantlets at a price of USh 1,300 per plantlet are about USh 1,430,000 per hectare (about US\$817). In our computations, we calculated the average annual cost of planting materials using a capital recovery factor for a 10% interest rate and an expected GM banana plant life cycle of 30 years. Furthermore, we assumed no price discount for the GM banana at the farm gate and no other costs of adoption.

The private incremental reversible benefits per hectare were used as the initial value for calculating the SIRBs. To obtain conservative estimates of SIRBs for Uganda, we assume that GM banana adoption follows a logistic function. We used this function to predict what the incremental benefits would be if the GM banana were adopted according to the logistic adoption function. We used an adoption ceiling rate of about 50% as a proxy for adopting any GM banana cultivar. This rate is based on

<sup>\*\*</sup> Source: Calculated by authors.

<sup>&</sup>lt;sup>a</sup> Tushemereirwe, Kashaija et al. (2003) recommend an average of about 1,100 plantlets per hectare. Currently, the cost of one no-GM tissue cultured plantlet is USh 1,000. Due to biosafety requirements, the cost of a GM banana plantlet may at least increase by 30% (USh 1,300).

<sup>8</sup> Following Griliches (1957) and Feder, Just, and Zilberman (1985), the adoption curve of a new technology is defined

 $P = \frac{K}{1 + ae^{-bt}}$ , where p is the percentage planted with GM banana in a given year, K is the ceiling rate (the long-term upper bound of adoption), a is the constant, related to the time when adoption starts, b is the speed of adoption, and t is the time variable. Parameters a and b were estimated using linear regression.

the predicted demand for nakitembe (a commonly grown cultivar) after effective insertion of genes with 60% resistance to both black Sigatoka and weevils, with supporting public investments in education, extension, and market-related infrastructure as estimated by Edmeades and Smale (2006). The adoption curve for an adoption ceiling of 50% and an estimated speed of adoption of 0.86 in linear form is p(t) = 3.2 - 0.86t. Figure 1 shows the assumed adoption curve.

We computed the SIRBs at time t (SIRB (t)), as the SIRBs at complete adoption times the adoption rate at time t, P(t), times the expected growth (or drift) at rate  $\alpha$ :  $SIRB(t) = SIRB \cdot P(t) \cdot e^{\alpha t}$ . The discounted sum of SIRBs,  $SIRB_{PV}$ , for Uganda over time is calculated as:

$$SIRB_{PV} = \int_0^\infty SIRB(t)e^{-(\mu-\alpha)t}dt, \qquad (2)$$

where  $\mu$  is the risk-adjusted discount rate and  $\alpha$  the drift rate of the geometric Brownian motion, explained in more detail below. The initial value for the calculation of the area for banana production is 1,670,000 hectares at full adoption.

Figure 1. GM banana adoption rate over time

# 60.0 50.0 40.0 30.0 20.0 10.0 0 5 10 15 20 Number of years

# Percent transgenic banana adopters with time

Source: Authors' own elaboration

In our analysis, we limit ourselves to the private incremental reversible benefits at the farm level, assuming all the rents from the new technology are captured by farmers. In the longer run, the rents will be distributed among farmers, the agents within the banana supply chain, and banana consumers. Additional secondary benefits such as improved food security and reduced vulnerability to external shocks may be generated through higher farm income among banana growers. Assessing such benefits would require the use of a general equilibrium model for Uganda and be beyond the scope of this study. Thus, the computed SIRBs are equal to the private incremental reversible benefits (PIRBs).

We also tried to identify the social incremental irreversible benefits (SIIBs) on a per hectare basis using information provided by Bagamba (2007). Most banana producers in Uganda do not use pesticides or fungicides to manage pests and diseases, as mentioned earlier. A small proportion (less than a quarter) of banana producers applies small amounts of pesticides.

In the literature on banana production, no shock has been reported as a result of epidemics of banana pests and diseases. If an epidemic did result as a consequence of introducing the GM banana, that would result in an irreversible social cost, even if the possibility existed for reversing the epidemic (Demont, Wesseler, and Tollens, 2005). Epidemics can be considered among irreversible costs. An epidemic is both a private and public irreversible cost. This is indirectly accounted for in the computed MISTICs—see Table 2, quadrants 3 and 4.

The hurdle rate,  $\beta/(\beta-1)$ , depends on the uncertainty related to the expected SIRBs. Secondary time series data on banana yield per hectare (UBOS, 2006b) were used to estimate the drift and variance of the geometric Brownian motion as a proxy for the drift and variance rate from gross margin time series data. The geometric Brownian motion  $U=(U_k(t),t\geq 0)$  is a continuous-time, continuous-state stochastic process in which the logarithm of the randomly varying quantity follows a Brownian motion:

$$U_k(t) = U_0 \exp\left[\left(\lambda - \frac{\sigma^2}{2}\right)t + \sigma W(t)\right]$$
; where  $W(t)$  is a Wiener process,  $U_0$  is the initial real random

number, t is the length of equally spaced intervals for all  $t \in [0, T]$ , and parameters  $\lambda$  and  $\sigma$  are constants.

The random variables  $\log \left(U_k/U_0\right) \equiv g_k\left(t\right)$  are independently and identically distributed with mean  $\left(\lambda - \sigma^2/2\right)t \equiv \alpha t$  ( $\alpha$  is the expected growth rate or drift) and variance  $\sigma^2 t$ , where  $k=0,1,\ldots,n$ . The maximum likelihood estimators for  $\alpha$  and  $\sigma^2$  were estimated as follows (see Campbell, Lo, and MacKinlay, 1997):

$$\alpha = \frac{1}{nt} \sum_{k=1}^{n} g_k(t), \tag{3}$$

$$\alpha = \frac{1}{nt} \sum_{k=1}^{n} g_k(t), \sigma^2 = \frac{1}{nt} \sum_{k=1}^{n} (g_k(t) - \alpha t)^2,$$
 (4)

where t, the length of intervals, was one year (t = 1), and n = 24 years (1980 through 2004). The estimated parameter values were ultimately used to derive hurdle rates for Uganda.

The different hurdle rates,  $\beta/(\beta-1)$ , were calculated as follows (see Dixit and Pindyck, 1994, pp. 147–52):

$$\beta = \frac{1}{2} - \frac{r - \delta}{\sigma^2} + \sqrt{\left[\frac{r - \delta}{\sigma^2} - \frac{1}{2}\right] + \frac{2r}{\sigma^2}} > 1,$$
 (5)

where r is the risk-free rate of return and  $\delta$  is the convenience yield defined as the difference between the risk-adjusted discount rate  $\mu$  and the drift rate  $\alpha$ ; i.e.,  $\alpha$ ,  $\mu \ge r$ , and  $\alpha$  and  $\sigma^2$  variance rate as before. From the Uganda Bureau of Statistics' data on total area and production of all types of bananas, we estimated the average yield per hectare. Since cooking bananas contribute 80% of total banana production in Uganda, this is a fairly good proxy for the yield of cooking bananas. We estimated a variance rate  $(\sigma)$  of 0.0328 and a drift rate  $(\alpha)$  of 0.0083 for the yearly difference change for all years from 1980 to 2004. Information about the risk-free rate of return and the risk-adjusted rate of return for farm household investments is rarely available and is difficult to calculate. Therefore, hurdle rates were calculated for different risk-free rates of return and risk-adjusted rates of return (0.04, 0.06, 0.08, 0.10, 0.12, and 0.14). Table 4 shows the computed annual SIRBs and MISTICs for a GM banana. The MISTICs are presented in total, on a per hectare, per household level, assuming 5,186,558 households as

of November 2002 (UBOS, 2006a), and per banana-growing farmer, assuming 1,500,000 banana-planting farm households (Kalyebara et al., 2006).

Table 4. Hurdle rates, average annual SIRBs and MISTICs per hectare of GM banana, per household, and per banana-growing farm household at different risk-free rates of return (r) and risk-adjusted rates of return  $(\mu)$ 

	SIRB (million \$) SIRB (\$/ha)	0.04 365 459	0.06 304	0.08 260	0.10	0.12	0.14
	SIRB (\$/ha)			260	227		
		459	200		226	200	179
			399	356	326	303	287
	Hurdle rate	1.0169	1.0104	1.0075	1.0059	1.0048	1.0041
	MISTIC (million \$)	359	301	258	225	199	178
0.00	MISTIC (\$/ha)	451	394	353	324	302	285
	MISTIC (\$/household)	69	58	50	43	38	34
	MISTIC (\$/farmer)	239	201	172	150	133	119
	Hurdle rate	1.3298	1.0405	1.0166	1.0103	1.0075	1.0058
	MISTIC (million \$)	274	293	256	224	198	178
0.04	MISTIC (\$/ha)	345	383	350	322	301	285
	MISTIC (\$/household)	53	56	49	43	38	34
	MISTIC (\$/farmer)	183	195	170	149	132	119
	Hurdle rate				1.1386	1.0355	1.0161
	MISTIC (million \$)				199	193	176
0.10	MISTIC (\$/ha)				286	293	282
	MISTIC (\$/household)				38	37	34
	MISTIC (\$/farmer)				132	129	118
)	.04	MISTIC (\$/ha) MISTIC (\$/household) MISTIC (\$/farmer)  Hurdle rate MISTIC (million \$)  MISTIC (\$/ha) MISTIC (\$/household) MISTIC (\$/farmer)  Hurdle rate MISTIC (million \$)  MISTIC (million \$)  MISTIC (\$/ha) MISTIC (\$/ha) MISTIC (\$/ha)	MISTIC (\$/ha) 451 MISTIC (\$/household) 69 MISTIC (\$/farmer) 239  Hurdle rate 1.3298 MISTIC (million \$) 274  .04 MISTIC (\$/ha) 345 MISTIC (\$/household) 53 MISTIC (\$/farmer) 183  Hurdle rate MISTIC (million \$) Hurdle rate MISTIC (\$/household) 53 MISTIC (\$/farmer) 183	MISTIC (\$/ha) 451 394 MISTIC (\$/household) 69 58 MISTIC (\$/farmer) 239 201  Hurdle rate 1.3298 1.0405 MISTIC (million \$) 274 293  .04 MISTIC (\$/ha) 345 383 MISTIC (\$/household) 53 56 MISTIC (\$/farmer) 183 195  Hurdle rate MISTIC (million \$)  Hurdle rate MISTIC (\$/household)  MISTIC (\$/ha) MISTIC (\$/ha) MISTIC (\$/ha) MISTIC (\$/ha) MISTIC (\$/ha) MISTIC (\$/household)	MISTIC (\$/ha) 451 394 353 MISTIC (\$/household) 69 58 50 MISTIC (\$/farmer) 239 201 172  Hurdle rate 1.3298 1.0405 1.0166 MISTIC (million \$) 274 293 256  .04 MISTIC (\$/ha) 345 383 350 MISTIC (\$/household) 53 56 49 MISTIC (\$/farmer) 183 195 170  Hurdle rate MISTIC (million \$)  Hurdle rate MISTIC (million \$)  MISTIC (\$/ha) MISTIC (\$/household)	MISTIC (\$/ha)	MISTIC (\$/ha)

Source: Authors' computation. Exchange rate \$1 = USh 1,750.

The results in Table 4 show that the SIRBs, as expected, decrease with an increase in the risk-adjusted rate of return. The estimated SIRBs range between US\$365 million and US\$179 million per year, or US\$459 and US\$287 per hectare per year, for the range of risk-adjusted discount rates that varied from 4% to 14%. The hurdle rates differ as the risk-free rate of return and risk-adjusted rate of return vary. For instance, at  $\mu = 0.1$  and r = 0.04, the hurdle rate is about 1.01. This means that on average every US\$1 of incremental social irreversible cost has to match with about 1.01 SIRBs to justify the immediate introduction of the GM banana. In general, the hurdle rates estimated in this paper are very

low compared with other estimates in the literature. This indicates that the irreversibility effect (Henry, 1974) is relatively small and much less important in comparison to other cases studies, where the hurdle rates range between 1.04 and 3.69 (Demont, Wesseler, and Tollens, 2004) and 1.03 and 5.6 (Wesseler, Scatasta, and Nillesen, 2007). We speculate that a potential explanation for the small value of irreversibility is that the statistical data from which we derived the MISTICs are not capturing production variability. Uganda's production data used to estimate the MISTICs are fairly smooth, in spite of observed biotic shocks in the 1990s and other years. Damage in a particular year may have been localized, yet heavy in those localized areas, so that national averages smooth out variations. Nevertheless, a smaller hurdle rate for Uganda implies that a delay in the introduction of a GM banana, for biosafety assessments or other reasons, affects benefits negatively.

The annual MISTICs decrease as well with an increase in the risk-adjusted rate of return and with an increase in the risk-free rate of return. At  $\mu=0.10$  and r=0.04, MISTICs are about US\$322 per hectare per year, or about US\$224 million per year. The MISTICs per banana-growing farm household and those per household indicate a large difference between the two groups. The MISTICs per farm household are more than three times larger than the MISTICs per household. As indicated previously, the MISTICs can be interpreted as the maximum willingness to pay for having a GM banana approved for planting in Uganda. Therefore, the difference in the MISTIC values between farm households and non-banana-producing households—both urban and rural—shows that in general the average banana-growing farm household may have a much larger interest than an average Ugandan household in having access to a GM banana even if the banana-growing household is concerned about the irreversible costs.

Non-banana-producing households in Uganda outnumber banana-producing households by about three to one. Therefore, the numbers presented here underscore the large magnitude of benefits foregone by Uganda, and thus reducing as much as possible the time needed to complete biosafety assessments becomes critical for the country. This statement supports scientists' opinion to not delay the introduction of disease-resistant bananas as the biological characteristics of the banana plant point to a lower risk as compared with other crops. As the banana plant is female sterile, an insignificant quantity of seed is produced, which is mainly sterile, and thus the probability of sexual reproduction or gene flow is insignificant. Therefore, the GM banana has a relatively safe environmental biosafety profile. An approach to assess whether benefit valuation is different between producer and consumer households is presented in the following section.

The specific GM banana that is going to undergo a confined field trial and posterior biosafety assessments is intended to control the fungal disease black Sigatoka. That disease affects mainly yield, but it also reduces bunch size. Consumers are able to observe the smaller bunch size but may not be able to associate the smaller size with the disease. The latter statement may also be true for other pests and diseases that primarily affect yield, such as nematodes, banana weevils, and other diseases. In contrast, the damage of diseases such as banana bacterial wilt is not readily apparent to consumers until after they purchase and consume the banana. In such a case, the consumer may be able to more readily associate damage with a disease. Thus, the value to consumers of a GM banana that is resistant to bacterial wilt may be higher. Consumers, in the case of bacterial wilt, may have a higher demand for a specific resistant variety. The increased demand may be high enough that consumers will pay a premium for a variety that guarantees an undamaged product. The latter statements, of course, become testable hypotheses for future work, and thus proposing methodologies to finely measure consumer willingness to pay for a GM banana becomes critical. We introduce one alternative methodology in the next section.

## 6. WILLINGNESS TO PAY FOR A GM BANANA

GM bananas are still currently a nontradable good in Uganda. A non-market valuation method (i.e., stated preference) can be used to determine the willingness to pay for a GM banana. Stated preference techniques involve asking respondents about their economic behavior, given a well-described artificial market scenario. Such techniques avoid some difficulties associated with revealed preference studies: they can assess demand for products that have poorly developed markets; they avoid issues of colinearity and low variability in explanatory variables; they are less demanding of research resources; and they can be experimentally designed to provide clear and easily interpretable results (Louviere, Hensher, and Swait, 2000).

Stated preferences are commonly elicited by using choice experiments. A choice experiment normally provides four pieces of information about a non-market good: (1) which attributes are significant determinants of the product value; (2) the relative rank of attributes among relevant populations; (3) the value of changing more than one attribute at once; and (4) the total economic value of the good (Bateman et al., 2003). A choice experiment is a highly structured method of data generation that relies on carefully designed tasks or experiments to reveal the factors that influence choice, but requires large sample sizes. The good is defined in terms of its attributes and the levels these attributes would take under different management scenarios. One of the attributes is the price, which enables estimation of the welfare measure, or value. Experimental design theory is used to construct profiles of such a good in terms of its attributes and attribute levels. Often, two or three alternative profiles are assembled in choice sets and presented to respondents, who are asked to state their preferred profile in each choice set (Louviere, Hensher, and Swait, 2000; Bennett and Blamey, 2001; Bateman et al., 2003).

The choice experiment method is one of the various choice models derived from random utility theory (e.g., McFadden, 1974; Ben-Akiva and Lerman, 1985). Lancaster (1966) proposed that consumers derive their satisfaction not from goods themselves but from the attributes they provide. Consumers try to maximize their utility by choosing the goods they think suit them best, all else equal. Chosen goods include products, services, and policies. The choice is subject to consumers' knowledge about competing options and individual constraints, such as income (Adamowicz et al., 1999; Bennett and Blamey, 2001; Bateman et al., 2003). The consumer n will choose option i over alternative j if  $U_{ni} > U_{nj}$ . Due to the random component, predictions cannot be made with certainty. Thus, the analysis becomes one of probabilistic choice. The probability that any particular respondent prefers a given option in the choice set to any other option can be expressed as the probability that the utility associated with a given option exceeds that associated with all other options:

$$P_{ni|C} = prob(U_{ni} \ge U_{nj}); i \ne j, \forall j \in C,$$
(6)

where the probability that individual n chooses alternative i in the choice set  $C(P_{ni|C})$  is proportionate with the probability that the utility  $U_{ni}$  is greater than the other alternatives  $U_{nj}$  in C. The equivalent indirect utility function, following Hanley, Wright, and Adamowicz (1998) and Adamowicz et al. (1999), can be stated as

$$U_{ni} = V_{ni}(X_{ni}) + e_{ni}, (7)$$

where  $U_{ni}$  is the total utility that individual n acquires from alternative i, which is decomposed into a deterministic component  $(V_{ni})$  and an error component  $(e_{ni})$ . The deterministic component is a function of a vector  $X_{ni}$ , consisting of attributes of a selected option and personal characteristics of the respondent n. The error  $e_i$  accounts for the entire unknown attributes and characteristics (Adamowicz et al., 1999) and is assumed to be independent of  $X_{ni}$  and to follow some predetermined distribution.

 $U_{ni}$  is a conditional indirect utility function (Adamowicz, Louviere, and Williams, 1994; Rolfe, Bennett, Louviere, 2002). Assuming the  $e_{ni}$  disturbances are Gumbel and are independently and

identically distributed (IID) with type-I extreme-value (Gumbel) distribution, the standard multinomial logit (MNL) model (McFadden, 1974) can be derived to estimate the parameters of  $U_{\it ni}$ , which takes the form

$$P_{in} = \frac{e^{\mu^{\beta X_{in}}}}{\sum_{j \in C} e^{\mu \beta X_{jn}}}.$$
 (8)

The scale parameter  $\mu$  is usually assumed to be equal to 1 (implying constant error variance) for the standard MNL model. This model is usually simple in constructing and analyzing data but imposes stringent behavioral restrictions on the data such as homogeneity with respect to individual preferences. This can lead to biased utility parameter estimates in cases where there is heterogeneity in preferences (Greene, 1977. Heterogeneity in preferences is a reasonable assumption for producers and consumers of GM crops.

Farmers have heterogeneous preferences regarding crop choices that depend on economic and socioeconomic factors and are well documented in the literature. Those preferences are likely to affect the willingness to pay (WTP) as well as the likelihood of adoption. The WTP of consumers in urban areas as well can be expected to differ and to depend on such differences as income, education, and occupation. Urban consumers may assign higher utility values to expensive, good quality, and more nutritious bananas. On the other hand, they may be more concerned about the future health risks of foods. Urban consumers have different sources of information and may differ in their capacity to utilize information. With genetic engineering remaining controversial, information on GM crops may influence consumers' willingness to pay for a GM banana. Difference in demand implies that homogeneity in tastes is unlikely to be the case for a GM banana.

To account for this heterogeneity a number of alternative models have been suggested. The commonly used ones are the heteroscedastic extreme value (HEV)<sup>9</sup> models, random parameter models,<sup>10</sup> mixed logit models (Greene and Hensher, 2003; Rigby and Burton, 2005), and latent class models (Swait, 1994; Louviere, Hensher, and Swait, 2000). The latent class models are suitable for capturing heterogeneity, while the basic multinomial logit models assume homogeneity in tastes and preferences. The random parameter logit model allows choice parameters to vary across individuals, but the latent segment (LS) approach assumes that choice parameters vary across segments of individuals. The LS approaches have been quite successful at identifying the sources of heterogeneity (Fredman and Emmelin, 2001; Kontoleon, 2003; Hu et al., 2004; Kontoleon and Yabe, 2006) at the segment level that are policy relevant. Kontoleon (2003) notes that unlike all other models that capture heterogeneity in consumer preferences, the LS model was found to be superior on policy grounds because it provides information on the viability of creating segregated food production, distribution networks, and markets, as well as an understanding of the welfare distributional impacts of alternative GM food policies.

In this study we apply an LS model similar to that of Swait (1994) to capture and account for heterogeneity among consumers (both farmers and urban consumers). The proposed model can generate the required parameters estimating respondents' willingness to pay to avoid or seek a level of an attribute or an alternative among members of the population. The use of the LS model developed by Swait (1994)

 $<sup>^9</sup>$  HEV models parameterize  $\mu$  with individual sociodemographic characteristics and relax the independent alternative assumption (IAA) property (see Louviere, Hensher, and Swait, 2000; Johnson, Banzhaf, and Desvousges, 2000). These models capture differences in respondent coherence, decision making, or interest in the activity (Breffle and Morey, 2000), but the approach does not examine how choice attributes differ across individuals with varying characteristics.

<sup>10</sup> Random parameter logit models allow all choice-specific parameters to vary across individuals but assume that the functional form and utility arguments are common. The models are not subject to the IAA property and perform better than conditional logit approaches (e.g., Breffle and Morey, 2000; Brownstone and Train, 1999; Revelt and Train, 1998). Some studies (e.g., Boxall and Adamowicz, 1999; Adamowicz et al., 1998) have raised concern that random parameter models are not appropriate for explaining the sources of heterogeneity.

and Boxall and Adamowicz (2002) allows one to identify the factors that drive consumers' willingness to pay and in particular the willingness (if any) to give up social benefits. The LS model identifies the sources of unobserved preference heterogeneity across individuals by revealing a finite number of latent segments of consumers that comprise individuals with tastes  $\beta_s$ , (s = 1,....,S) unique to that segment or that have different underlying variances (Swait, 1994; Louviere, Hensher, and Swait, 2000). The model postulates an indirect utility function of the form

$$U_{ni/s} = \beta_s X_{ni} + e_{ni/s}, \tag{9}$$

where *i* is the alternative chosen by individual *n* belonging to segment *s*, in which  $e_{ni/s}$  are IID and of extreme-value type-I (or Gumbel) distribution.

The probability of choosing alternative i by the nth individual in segment s is then given by

$$P_{in/s} = \frac{e^{\mu_s(\beta_s X_{in})}}{\sum_{j \in C} e^{\mu_s(\beta_s X_{jn})}},$$
(10)

where  $\mu$  is a scale factor that captures difference in variances, and the probability of an individual n being in segment s is

$$W_{ns} = \frac{e^{\lambda(\alpha_s Z_n)}}{\sum_{s=1}^{S} e^{\lambda(\alpha_s Z_n)}},$$
(11)

where  $Z_n$  is a column vector of observed and unobserved characteristics of the individual and  $\alpha_s$  is the row vector of the corresponding parameters. Bringing equations 10 and 11 together, we construct an unconditional probability of individual n belonging to segment s choosing alternative i:

$$P_{isn} = \sum_{s=1}^{S} P_{in/s} W_{ns} = \sum_{s=1}^{S} \left[ \frac{e^{\mu_s(\beta_s X_{in})}}{\sum_{j \in C} e^{\mu_s(\beta_s X_{jn})}} \right] \cdot \left[ \frac{e^{\lambda(\alpha_s Z_n)}}{\sum_{s=1}^{S} e^{\lambda(\alpha_s Z_n)}} \right].$$
(12)

The number of segments in the LS model can be endogenously determined with the utility coefficients using various information criteria (Boxall and Adamowicz, 2002; Andrews and Currim, 2003). Thus, instead of arbitrarily dividing consumers into different segments based on observable similarities (e.g., sociodemographic characteristics) and estimating respective preferences for each individual class, the LS model uses the information derived from individual choices.

#### The WTP Welfare Measures in the Latent Segment Model

Once models have been developed and the required parameters generated, respondents' willingness to pay to give up social benefits of a GM banana can be estimated. The LS model allows us to calculate WTP welfare measures for each respondent within a segment. Deriving welfare measures under the LS models is done in two steps. First, policy impacts at the segment level are identified by calculating WTP welfare measures for each segment. Second, the standard aggregate procedure (that assumes homogeneous preference) is corrected for heterogeneity. That is done by computing the weighted sum of segment-specific welfare measures. The weights are the estimated individual segment membership probabilities (Boxall and Adamowicz, 2002). The individual segment WTP can finally be aggregated to estimate WTP welfare measures for the whole population.

Following Hanemann (1983, 1984), the general WTP welfare measure can be calculated as

$$CV_n = \frac{1}{\theta} \left[ \ln \left( \sum_{i \in C} e^{\beta X_i^0} \right) - \ln \left( \sum_{i \in C} e^{\beta X_i^1} \right) \right], \tag{13}$$

where the compensating variation  $CV_n$  is the amount of money that one would have to give the individual n after the change has occurred in order to remain as well off as before;  $\theta$  is the marginal utility of money and is the coefficient of the price attribute;  $\beta X_i^0$  represents the individual's utility at the initial level (i.e., prechange state); and  $\beta X_i^1$  is the utility of the alternative level (i.e., postchange state) following changes in attributes in X.

The final marginal WTP welfare measure can be derived by, first, integrating the welfare effects across the different segments,

$$CV_{n|s} = \frac{1}{\theta_s} \left[ \ln \left( \sum_{I \in C} e^{\beta_s X_I^0} \right) - \ln \left( \sum_{I \in C} e^{\beta_s X_I^1} \right) \right], \tag{14}$$

and, second, by calculating the weighted sum of the segment membership:

$$CV_{n|s}^{t} = \sum_{s=1}^{S} W_{ns} \left\{ \frac{1}{\theta_{s}} \left[ \ln \left( \sum_{I \in C} e^{\beta_{s} X_{I}^{0}} \right) - \ln \left( \sum_{I \in C} e^{\beta_{s} X_{I}^{1}} \right) \right] \right\}, \tag{15}$$

where  $W_{ns}$  is the probability of an individual n being in segment s.

# Suggested Sampling Procedures and Data Collection for a GM Banana in Uganda

The WTP study can be conducted in different regions in Uganda where cooking bananas (matooke) are produced and consumed, including urban areas that are net consumers of bananas. Implementation of this procedure allows for the capture of heterogeneity in preferences across different population segments. The survey sample can be drawn using a multistage sampling procedure from the major banana-producing regions in eastern, central, and southwestern Uganda. The 2002 Uganda census indicates that only 12.3% of the population resides in urban areas. Those areas include cities, municipalities, and town councils (UBOS, 2006a). Every day, 65% of urban consumers in Uganda have a meal of matooke (Clarke, 2003). More than three-quarters of bananas produced are consumed on the farm (NARO, 2001), implying that a large proportion of matooke production is for subsistence. Respondents should therefore be stratified into rural consumers (producers and consumers) and urban consumers of matooke and receive extra information about the GM banana.

Studies have revealed that willingness to purchase GM food products increases especially for GM foods with tangible consumer benefits (Loureiro and Hine, 2002; Li et al., 2003; Loureiro and Bugbee, 2005; Onyango and Govindasamy, 2005). All choice sets should therefore incorporate several attributes of the GM banana, including the incremental social benefits. In addition, respondents will be asked about their awareness of, perceptions of, and attitudes toward biotechnology and the biosafety of genetically modified products. Socio-demographic characteristics will also be captured. This allows one to identify the factors that influence the willingness to pay and, in particular, the willingness (if any) to give up social benefits.

The primary sampling unit (PSU) should be the subcounty for rural areas and the municipality/town council for urban areas. A total number of 12 PSUs will be selected—three-quarters in

rural areas, and a quarter in urban areas. This is based on the distribution of the Ugandan population<sup>11</sup> as reported in the 2002 census. The secondary sampling unit is defined at the village level, including 20 randomly selected households with 40 households per PSU. A sample size of about 480 households is needed for the suggested attributes and attribute levels. Table 5 shows attributes (consumption and production) and levels for choice sets for the GM banana. <sup>12</sup> Tables 6 and 7 show examples of possible choice scenarios.

Table 5. Attributes, their definitions, and levels for choice sets

Attribute <sup>a</sup>	Definition	Attribute level
Bunch size	The average weight in kilograms of a banana bunch at harvest, categorized into small, medium, and large	5 to 15 (small) 16 to 25 (medium) 26+ (large)
Extra benefit	The estimated extra monetary benefit per hectare in Ugandan shillings that would result if a gene were inserted in cooking banana (matooke) planting materials in order to improve resistance to pests and disease (e.g., banana bacterial wilt or black Sigatoka)	USh 0 USh 60,000 USh 120,000
Technology	The technology used to produce the banana planting material	Tissue culture only Tissue culture + GM
Price	Hypothetical percent change in price of a bunch of bananas	-30 -15, 0, +15, +30 +40

Source: <sup>a</sup> Attributes were derived from focus group interviews with farmers, as well as from previous work on banana attributes in Uganda. For more information, several studies can be consulted: Smale and Tushemereirwe (2007); Edmeades (2003); Nowakunda et al. (2000); Karamura et al. (1998); Gold et al. (1998).

Table 6. Example of a choice scenario with a benefit presented

Attribute	Option A	Option B	Option C
Bunch size (Kg)	5–15	26+	
Extra benefit to farmer per hectare (USh)	0	120,000	I would not buy any of those mentioned banana
Technology	Tissue culture only	Tissue culture + GM	those mentioned banana
Price (% change)	15	0	
Indicate your choice by a tick in any one box			

Source: Author's own elaboration

<sup>11</sup> Less than a quarter of the population is in urban areas, but the selected regions contribute over 90%.

<sup>&</sup>lt;sup>12</sup> The perceived consumption (cooking) quality of the GM banana (matooke) is expected to be akin to that of non-GM matooke—that is, with good taste, soft texture, and yellow color.

Table 7. Example of a choice scenario with no benefit presented

Attribute	Option A	Option B	Option C
Bunch size (Kg)	16-25	5-15	
Extra benefit to farmer per hectare (USh)	0	0	I would not buy any of those mentioned banana
Technology	Tissue culture +GM	Tissue culture only	
Price (% change)	40	0	
Indicate your choice by a tick in any one box			

Source: Author's own elaboration

# **Integrating MISTICs and WTP**

The MISTICs associated with the immediate introduction of a GM banana can be compared with the estimated WTP values for the GM banana for different scenarios. The MISTICs provide a threshold value for the immediate release of a GM banana. The WTP shows how much money consumers are willing to pay to accept or avoid the introduction of a GM banana. Therefore, if the WTP for not having a GM banana introduced is greater than the MISTICs, then arguments can be advanced for delaying the introduction of the GM banana. If the WTP for not having a GM banana is positive but below the MISTICs, then very good economic arguments exist for not further delaying the introduction of the GM banana. An advantage of estimating the WTP is that the results of the study will allow identification of the differences between segments of producers and consumers that can help to formulate national policies. We expect that the parallel (joint) estimation of the MISTICs, the SIRBs, and the WTP will become an invaluable technique in the socioeconomic impact assessment of genetically modified biotechnologies and other technologies.

# 7. SUMMARY, CONCLUSIONS, AND RECOMMENDATIONS

In this study we have presented an approach for considering concerns about genetically modified crops within a socioeconomic analysis of GM crops. The study proposes a framework for the GM banana. The approach presented relates the economic benefits to consumers' concerns. We propose to compare the maximum incremental social tolerable irreversible costs (MISTICs) with the willingness to pay for the technology. The suggested procedure considers not only the possible long-term effects but also the short-term benefits and costs of the technology. The comparison provides information on whether the introduction of a GM banana poses a socioeconomic problem from the consumer's point of view.

We calculated the MISTICs associated with the adoption of a GM banana in Uganda. The MISTICs were presented for different risk-free and risk-adjusted rates of return. The preliminary results show the MISTICs to be between approximately US\$176 million and US\$359 million per year, or between US\$282 and US\$451 per hectare per year. In the scenario with a risk-adjusted rate of return of 12% and a risk-free rate of interest of 4%, which we consider to be a reasonable scenario, the annual MISTICs per household are about US\$38. This result can be interpreted as follows: the immediate release of the GM banana should be postponed or abandoned only if the average household is willing to give up more than US\$38 per year for not having such a banana introduced.

In the case where approval of the GM banana is delayed due to missing regulatory procedures and protocols, Uganda will forego potential benefits (social incremental reversible benefits, or SIRBs) in the approximate range of US\$179 million to US\$365 million per year. This foregone benefit can be an indicator of how much Uganda can pay to compensate for potential damages. Additionally, the SIRBs provide a clue about the maximum costs farmers would endure in order to comply with biosafety regulations, including the cost of implementing coexistence policies and after deducting planting costs of US\$101 per hectare. In a reasonable scenario, for instance, the average SIRBs total about US\$303 per hectare. Adopters of the GM banana would not be willing to pay more than US\$200 per hectare per year in transaction costs—i.e., costs to comply with biosafety regulations, R&D costs, and technology transfer costs. If the average WTP per hectare of a banana-growing household is below the MISTICs but biosafety regulators are inclined to implement biosafety regulations to address concerns of consumers with a high WTP for not having the GM banana, those additional costs should not exceed US\$200 on average per year per hectare of GM banana. Assuming a maximum of 541,530 hectares that may be planted in GM banana in Uganda, this implies that the maximum total costs to bring the GM banana to Ugandan producers cannot exceed US\$108 million. Otherwise, the GM banana is not a viable alternative.

One of the limitations of this analysis, however, is that we have not included the transaction costs that might be involved between the technology developers and the end users, including R&D costs, compliance with biosafety regulatory costs, and technology fees. Such costs can be substantial and are one of the major obstacles to technology dissemination in developing countries such as Uganda (Brenner, 2004). The problem is not limited to GM technology but to embodied technologies in general. Adding such costs will reduce the SIRBs. Again, they should on average not be more than the SIRBs per hectare, and should be even less if biosafety regulatory costs at the farm level are added.

The approach for assessing the WTP contributes to the knowledge of understanding the extent of concerns (e.g., consumer attitudes toward biosafety risk) about GMOs in a heterogeneous banana industry. The approach used in this study illustrates a framework that can be used to identify the different consumer segments within the population—their size, their characteristics, and how they differ. In this regard, the biosafety regulators can, first, know what would be the impact of introducing a GM banana to consumers with heterogeneous preferences and, second, have a rough estimate of how much would be required to compensate a given population segment if a GM banana were introduced. Finally, policymakers would have explicit information about the risk margins for introducing a GM banana (or other crop), which contributes to better-informed debates and a more transparent policymaking process.

21

<sup>&</sup>lt;sup>13</sup> As technology fees charged by innovators are used to recover R&D costs and biosafety costs, it is imperative to include such costs as net costs to society to avoid double-counting.

# Implications for Decision Making on Biotechnology and Biosafety Regulations

The analyses in this paper demonstrate the economic value and the effect of the foregone benefits as a result of waiting to release a GM banana. The results illustrate several implications to numerous stakeholders. First, the calculation of the MISTICs considers explicitly possible long-term effects of GM banana. The results indicate that with each year of delay in the introduction of a GM banana, Uganda loses about US\$179 million to US\$365 million to all households in Uganda. The MISTICs are on the order of about US\$176 million or more. Only if the real average annual irreversible costs of planting a GM banana would be as high, or higher than, the irreversible benefits, should the release be delayed. We have found no evidence yet that this will be the case. Given the potential and significant economic benefits from the introduction of a GM banana, NARO has to work harder to push the GM banana through the biosafety protocols as promptly and efficiently as possible.

Second, the results may not convince all stakeholders. There is an urgent need for NARO and other R&D institutions to conduct research to understand how consumers feel about GM banana biosafety risks and the potential challenges for marketing the product. We have illustrated how one could implement such a study of consumer perception. Further research on implementing the assessment procedure that this paper describes has been initiated by the authors in Uganda. The experience gathered from the field research in Uganda can serve to establish a feasible and practical system for consumer assessments as part of the approval process for GM crops if indeed socioeconomics become part of the decision-making process for approval before commercialization.

Third, our findings indicate that a banana-growing household may have a much (three times) larger interest in having access to a GM banana than an average Ugandan household. This can be explained by the great losses experienced by farm households due to the prevailing banana constraints. The losses caused by banana constraints, therefore, make the opportunity cost to farmers of not using the GM banana technology extremely high. This implies that a farm household would naturally benefit disproportionately from a GM banana technology that is likely to ensure a return to sustainable production.

The challenge decision makers (regulators) face is to develop a regulatory process that will ensure a high degree of safety without imposing stringent biosafety regulations on the development and accessibility of the technology. Unfortunately, such a process may mean substantial time will pass before the product can be accessed by farm households. The reasons for that may be, first, that approval of the GM banana in Uganda has followed a sequential protocol of biosafety regulation where each stage demands an application to and approval by the National Biosafety Committee, and second, that a fungus-(or bacteria-) resistant banana is a novel crop in terms of the gene, the transformation process, and the crop. There is very little regulatory experience elsewhere with any of these components. There may be additional delays in terms of regulatory time for approval as a result of the likely regulatory procedures and protocols to prove the safety of the gene, the transformation protocol, and the final novel product. The implication of this extended procedure will be a (likely) higher cost of compliance with the biosafety assessments. A longer time, a greater effort, and thus a greater cost than is necessary to prove safety in the biosafety regulatory analysis of a GM banana may presumably result in failure to access the expected benefits of the GM banana.

Biosafety regulatory assessment, and its posterior analysis, has to overcome the observed tendency of most regulatory processes globally of avoiding committing regulatory errors during decision making and particularly stacking the odds in favor of not approving technologies that are safe against approving a technology that is not safe. In essence, decisions made by most regulatory bodies tend to be more precautionary than warranted. To ensure a more balanced approach to decision making, the literature suggests consideration of all benefits and costs—including opportunity and irreversible—supporting regulatory decision making. This paper proposes one alternative in this line of reasoning.

Lastly, the approach used here highlights how one can evaluate the socioeconomic aspects of GM crops in general. To those stakeholders who are pessimistic about such technologies, it shows how much benefits are foregone as a result of a delayed release of a given technology. We have also indicated how

one can consider long-term irreversible effects and how one might assess consumer attitudes toward GM crops. The approach can therefore be adapted to new GM crops requiring biosafety assessments prior to commercialization and can help to overcome one of the problems of establishing a biosafety system for Uganda. In particular, NARO may want to institutionalize the approach suggested in this paper and build a system that allows for conducting similar analyses of other GM crops. Banana is one of the staple crops in Uganda and is integral to ensuring the livelihood of households in the country. Addressing and resolving the binding constraints on banana production in Uganda therefore will contribute to ongoing poverty alleviation efforts and compliance with the Millennium Development Goals and overall sustainable social and economic development.

## **REFERENCES**

- Adamowicz, W., P. C. Boxall, J. Louviere, Y. Swait, and N. Williams. 1999. Stated preference methods for valuing environmental amenities. In *Valuing environmental preferences*, edited by I. J. Bateman and K. G. Willis. Oxford: Oxford University Press.
- Adamowicz, W., P. C. Boxall, M. Williams, and J. Louviere. 1998. Stated preference approaches for measuring passive use values: Choice experiments and contingent valuation. *American Journal of Agricultural Economics* 80 (1): 64–75.
- Adamowicz, W., J. Louviere, and M. Williams. 1994. Combining revealed and stated preference methods for valuing environmental amenities. *Journal of Environmental Economics and Management* 26: 271–92.
- Andrews, R. L., and I. S. Currim. 2003. A comparison of segment retention criteria for finite mixture logit models. *Journal of Marketing Research* 40: 235–43.
- Arrow, K. J. and A.C.Fisher. 1974. Environmental preservation, uncertainty and irreversibility. *Quarterly Journal of Economics* 88:312-319.
- Atanassov, A., A. Bahieldin, J. Brink, M. Burachik, J. I. Cohen, V. Dhawan, R. V. Ebora, J. Falck-Zepeda, L. Herrera-Estrella, J. Komen, F. C. Low, E. Omaliko, B. Odhiambo, H. Quemada, Y. Peng, M. J. Sampaio, I. Sithole-Niang, A. Sittenfeld, M. Smale, Sutrisno, R. Valyasevi, Y. Zafar, and P. Zambrano. 2004. To reach the poor: Results from the ISNAR-IFPRI Next Harvest Study on Genetically Modified Crops, Public Research, and Policy Implications. EPTD Discussion Paper 116, International Food Policy Research Institute, Washington, D.C.
- Bachelier, L. 1900. *Theory of speculation*. Paris: Gauthier-Villars. Translated by A. J. Boness and reprinted in P. Cootner, *The random character of stock market prices*. Cambridge: MIT Press, 1964.
- Bagamba, F. 2007. Market access and agricultural production: The case of banana production in Uganda. Doctoral thesis, Wageningen University, Netherlands.
- Bateman, I. J., R. T. Carson, B. Day, W. M. Hanemann, N. Hanley, T. Hett, M. Jones-Lee, G. Loomes, S. Mourato, E. Ozdemiroglu, D. W. Pearce, R. Sugden, and T. Swanson. 2003. *Guidelines for the use of stated preference techniques for the valuation of preferences for non-market goods*. Cheltenham, UK: Elgar.
- Beckmann, V., C. Soregaroli, and J. Wesseler. 2006a. Governing the co-existence of GM crops: Ex-ante regulation and ex-post liability under uncertainty and irreversibility. Institutional Change in Agriculture and Natural Resources (ICAR) Discussion Paper 12/2006, Humboldt University, Berlin.
- ——. 2006b. Co-existence rules and regulations in the European Union. *American Journal of Agricultural Economics* 88 (5): 1193–99.
- Ben-Akiva, M., and S. Lerman. 1985. *Discrete choice analysis: The theory and application to predict travel demand.* Cambridge: MIT Press.
- Bennett, J. J., and R. K. Blamey. 2001. *The choice of modeling approach to environmental valuation*. Cheltenham, UK: Elgar.
- Bernstein, P. L. 1992. Capital ideas: The improbable origins of modern wall street. New York: Free Press.
- Boxall, P. C., and W. L. Adamowicz. 1999. Understanding heterogeneous preferences in random utility models: The use of latent class analysis. Staff Paper 99-02, Department of Rural Economy, University of Alberta, Edmonton.
- ———. 2002. Understanding heterogeneous preferences in random utility models: A latent class approach. Environmental and Resource Economics 23: 421–46.
- Breffle, W. S., and E. R. Morey. 2000. Investigating preference heterogeneity in a repeated discrete-choice recreation demand model of Atlantic salmon fishing. *Marine Resource Economics* 15 (1): 1–20.

- Brenner, C. 2004. Telling transgenic technology tales: Lessons from the Agricultural Biotechnology Support Project (ABSP) experience. ISAAA (International Service for the Acquisition of Agri-biotech Applications) Brief 31, Ithaca, N.Y.
- Brownstone, D., and K. Train. 1999. Forecasting new product penetration with flexible substitution patterns. *Journal of Econometrics* 89 (1–2): 109–29.
- Campbell, J. Y., A. W. Lo, and A. C. MacKinlay. 1997. *The econometrics of financial markets*. Princeton, N.J.: Princeton University Press.
- Clarke, T. 2003. Banana lab opens in Uganda: Genetic modification of clonal crop could soon follow. *Nature News*, August 22. http://www.bioedonline.org/news/news.cfm?art=430.
- Cox, D. and H. Miller. 1965. The theory of stochastic processes. London: Chapman and Hall.
- Craenen, K. 1998. *Black Sigatoka disease of banana and plantain: A Reference Manual.* Ibadan: International Institute of Tropical Agriculture.
- Demont, M., J. Wesseler, and E. Tollens. 2004. Biodiversity versus transgenic sugar beet: The one Euro question. *European Review of Agricultural Economics* 31 (1): 1–18.
- ———. 2005. Irreversible costs and benefits of transgenic crops: what are they? In *Environmental costs and benefits of transgenic crops*, edited by J. Wesseler, pp. 113-122. Dordrecht: Springer.
- Dixit, A., and R. S. Pindyck. 1994. Investment under uncertainty. Princeton, N.J.: Princeton University Press.
- Edmeades, S. 2003. Variety choice and attribute trade-offs within the framework of agricultural household models: The case of bananas in Uganda. Doctoral dissertation, North Carolina State University, Raleigh.
- Edmeades, S., and M. Smale. 2006. A trait-based model of the potential demand for a genetically engineered food crop in a developing economy. *Agricultural Economics* 35: 351–61.
- Falck-Zepeda, J. B. 2006. Coexistence, genetically modified biotechnologies, and biosafety: Implications for developing countries. *American Journal of Agricultural Economics* 88 (5): 1200–08.
- Feder, G., R. E. Just, and D. Zilberman. 1985. Adoption of agricultural innovations in developing countries: A survey. *Economic Development and Cultural Change* 33: 255–98.
- Food and Agriculture Organization (FAO). 2001. Biotechnology and banana production. Committee on Commodity Problems Intergovernmental Group on Bananas and Tropical Fruits, San Jose, Costa Rica.
- Fredman, P., and L. Emmelin. 2001. Wilderness purism, willingness to pay, and management preferences: A study of Swedish mountain tourists. *Tourism Economics* 7 (1): 5–20.
- Gold, C. S. 1998. Banana weevil: Ecology pest status and prospects for integrated control with emphasis on East Africa. In S. K Saini, ed. *Proceedings of a symposium on biological control in tropical habitats: Third International Conference on Tropical Entomology*, edited by S. K. Saini, pp. 49–74. Nairobi: International Centre of Insect Physiology and Ecology(ICIPE)
- ——. 2000. Biology and integrated pest management of banana weevil, *Cosmopolites*
- sordidus (Germar). In Advancing banana and plantain R&D in Asia and the Pacific, vol. 10, edited by A. B. Molina, V. N. Roa, and M. A. G. Maghuyop, pp. 28–33. International Network for the Improvement of Banana and Plantain Asia and the Pacific Network (INIBAP-ASPNET),
- Gold, C. S., G. Night, A. Abera, and P. R. Speijer. 1998. Hot-water treatment for the control of the banana weevil, *Cosmopolites sordidus* Germar (*Coleoptera: Curculionidae*), in Uganda. *African Entomology* 6 (2): 215–21.
- Gold, C. S., J. E. Pena, and E. B. Karamura. 2001. Biology and integrated pest management for the banana weevil, Cosmopolites sordidus (Germar) (Coleoptera: Curculionidae). Integrated Pest Management Reviews 6 (2): 79–155
- Government of Uganda (GOU). 2002a *Guidelines on biosafety in biotechnology for Uganda*. Kampala, Uganda: Uganda National Council of Science and Technology.

- ———. 2002b. *National biodiversity strategy and action plan: National Environment Management Authority (NEMA)*, 1st ed. The Convention on Biological Diversity around the World (CBD) website: http://www.biodiv.org/doc/world/ug/ug-nbsap-01-en.doc.
- ——. 2004. Draft of national biotechnology and biosafety policy. Uganda National Council of Science and Technology, Kampala, Uganda.
- ——. 2005. The Uganda biosafety bill. Kampala, Uganda.
- Greene, W. 1997. Econometric analysis. New York: Prentice Hall.
- Greene, W., and D. Hensher. 2003. A latent class model for discrete choice analysis: Contrasts with mixed logit. *Transportation Research, Part B* 37: 681–98.
- Griliches, Z. 1957. Hybrid Corn: An exploration in the economics of technological
- change. *Econometrica* 25 (4): 501-522.
- Haneman, W. M. 1983. Marginal welfare measures for discrete choice models. *Economics Letters* 13: 129–36.
- ——. 1984. Applied welfare analysis with quantitative response models. Working Paper no. 241. Berkeley: University of California.
- Hanley, N., R. E. Wright, and W. L. Adamowicz. 1998. Using choice experiments to value the environment. *Environmental and Resource Economics* 11 (3–4): 413–28.
- Henry, C. 1974. Investment decision under uncertainty: The irreversibility effect. *American Economic Review* 64: 1006–12.
- Hu, W., A. Hünnemeyer, M. Veeman, W. Adamowicz, and L. Scrivatava. 2004. Trading off health, environmental, and genetic modification attributes in food. *European Review of Agricultural Economics* 31 (3): 389–408.
- Jaffe, G. 2006. Comparative analysis of the national biosafety regulatory systems in East Africa. EPTD Discussion Paper 146, International Food Policy Research Institute, Washington, D.C.
- Johnson, F. R., M. R. Banzhaf, and W. H. Desvousges. 2000. Willingness to pay for improved respiratory and cardiovascular health: A multiple-format, stated preference approach. *Health Economics* 9: 295–317.
- Just, R. E., J. M. Alston, and D. Zilberman. 2006. What have we learned and where do we go from here? In *Regulating agricultural biotechnology: Economics and policy*, edited by R. E. Just, J. M. Alston, and D. Zilberman, pp. 701–22. New York: Springer.
- Kalaitzandonakes, N., J. Alston, and K. Bradford. 2007. Compliance costs for regulatory approval of new Biotech crops. *Nature Biotechnology*, 25(5): 509-11.
- Kalyebara, M. R., P. E., Ragama, G. H. Kagezi, J. Kubiriba, F. Bagamba, K. C. Nankinga, and W. K. Tushemereirwe. 2006. Economic importance of the banana bacterial wilt in Uganda. *African Crop Science Journal* 14 (2): 93–105.
- Karamura, D. A. 1998. Numerical taxonomic studies of the East African highland bananas (Musa AAA–East Africa) in Uganda. Doctoral thesis, University of Reading, United Kingdom.
- Karamura, E., E. Frison, D. Karamura, and S. Sharrock. 1998. Banana production systems in eastern and southern Africa. In *Bananas and food security*, edited by C. Picq, E. Foure, and E. A. Frison. International Symposium, Cameroon, November 10–14.
- Kendall, H. W., R. Beachy, T. Eisner, F. Gould, R. Herdt, P. H. Raven, J. S. Schell, and M. S. Swaminathan. 1995. Bioengineering of crops. Report of the World Bank Panel on Transgenic Crops. World Bank, Washington, D.C.

- Kikulwe, M. E., K. Nowakunda, M. S. R. Byabachwezi, J. M. Nkuba, J. Namaganda, D. Talengera, E. Katungi, and W. K. Tushemereirwe. 2007. Development and dissemination of improved banana cultivars and management practices in Uganda and Tanzania. In *An economic assessment of banana genetic improvement and innovation in the Lake Victoria region of Uganda and Tanzania*, edited by M. Smale and W. K. Tushemereirwe. Research report no. 155 ch. 4, International Food Policy Research Institute, Washington, D.C.
- Knight, J.A., D. W. Mather, D. K. Holdsworth, and D. F. Ermen. 2007. Acceptances of GM food an experiment in six countries. Nature Biotechnology 25(5): 507-508.
- Kontoleon, A. 2003. Accounting for consumer heterogeneity in preferences over GM foods: An application of the latent market segmentation model. Paper presented at Fourth BIOECON Workshop on Economics and Biodiversity Conservation, Venice, Italy, August, 28<sup>th</sup> 29<sup>th</sup>
- Kontoleon, A., and M. Yabe. 2006. Market segmentation analysis of preferences for GM derived animal foods in the UK. *Journal of Agricultural and Food Industrial Organization* 4 (1): 8.
- Lancaster, K. 1966. A new approach to consumer theory. Journal of Political
- Economy. 84: 132-157
- Li, Q., K. R. Curtis, J. J. McCluskey, and T. I. Wahl. 2003. Consumer attitudes towards genetically modified foods in Beijing, China. *AgBioForum* 5: 145–52.
- Loureiro, M. L., and M. Bugbee. 2005. Enhanced GM foods: Are consumers ready to pay for the potential benefits of biotechnology? *Journal of Consumer Affairs* 39 (1): 52.
- Loureiro, M. L., and S. Hine. 2002. Discovering niche markets: A comparison of consumer willingness to pay for local (Colorado grown), organic, and GMO-free products. *Journal of Agricultural and Applied Economics* 34 (3): 477–87.
- Louviere, J. J., D. A. Hensher, and J. D. Swait. 2000. *Stated choice methods: Analysis and application*. Cambridge: Cambridge University Press.
- McFadden, D. 1974. Conditional logit analysis of qualitative choice behaviour. In *Frontiers in Econometrics*, edited by P. Zarembka. New York: Academic Press.
- Modgil, M., K. Mahajan, S. K. Chakrabarti 1, D. R. Sharma, and R. C. Sobti. 2005. Molecular analysis of genetic stability in micropropagated apple rootstock MM106. *Scientia Horticulturae* 104: 151–60.
- Nampala, P., C. Mugoya, and T. Sengooba. 2005. Biosafety regulatory system in Uganda. Presentation to Program for Biosafety Systems Roundtable, Entebbe, Uganda.
- National Agricultural Research Organization (NARO). 2001. Multilocational banana germplasm evaluation trials: Third report. Unpublished report, NARO Secretariat, Entebbe, Uganda.
- Nowakunda, K. 2001. Determination of consumer acceptability of introduced bananas. Unpublished master's thesis, Makerere University, Uganda.
- Nowakunda, K., P. R. Rubaihayo, M. A. Ameny, and W. K. Tushemereirwe. 2000. Consumer acceptability of introduced bananas in Uganda. *InfoMusa* 9 (2): 22–25.
- Onyango, B., and R. Govindasamy. 2005. Consumer willingness to pay for GM food benefits: Pay-off or empty promise? Implications for the food industry. *Choices* 20 (4): 223–26.
- Oxford Policy Management (OPM). 2005. Evaluation of the Plan for the Modernization of Agriculture: Main report. http://www.pma.go.ug/pmauploads/Main%20PMA%20evaluation%20final%20report.pdf.
- Revelt, D., and K. Train. 1998. Mixed logit with repeated choices: Households' choice of appliance efficiency level. *Review of Economics and Statistics* 53: 647–57.
- Rigby, D., and M. Burton. 2005. Preference heterogeneity and GM food in the UK. European Review of Agricultural Economics 32 (2): 269–88.

- Rolfe, J., J. Bennett, and J. Louviere. 2002. Stated values and reminders of substitute goods: Testing for framing effects with choice modeling. *Australian Journal of Agricultural and Resource Economics* 46 (1): 1–20.
- Rubaihayo, P. R. 1991. Banana-based cropping systems research: A report on rapid appraisal survey of banana production in Uganda. Research Bulletin 1. Department of Crop Science, Makerere University, Kampala, Uganda.
- Rubaihayo, P. R. and C. Gold. 1993. Banana-based cropping systems research: A report on rapid appraisal survey of banana production in Uganda. Research Bulletin 2. Department of Crop Science, Makerere University, Kampala, Uganda.
- Scatasta, S., J. Wesseler, and M. Demont. 2006. Irreversibility, uncertainty, and the adoption of transgenic crops: Experiences from applications to Ht sugar beet, Ht corn, and Bt corn. In *Regulating agricultural biotechnology: Economics and policy*, edited by R. E. Just, J. M. Alston, and D. Zilberman, pp. 327–52. New York: Springer.
- Secretariat of the Convention on Biological Diversity. 2000. Cartagena protocol on biosafety to the Convention on Biological Diversity: Text and annexes. Montreal: Secretariat of the Convention on Biological Diversity.
- Smale, M., and W. K. Tushemereirwe, eds. 2007. *An economic assessment of banana genetic improvement and innovation in the Lake Victoria region of Uganda and Tanzania*. Research report no. 155, International Food Policy Research Institute, Washington, D.C.
- Ssebuliba, R.N. 2001. Fertility in East African highland bananas. Doctoral thesis, Makerere University, Uganda.
- Ssebuliba, R., D. Talengera, D. Makumbi, P. Namanya, A. Tenkouano, W. Tushemereirwe, and M. Pillay. 2006. Reproductive efficiency and breeding potential of East African highland (Musa AAA–EA) bananas. *Field Crops Research* 95: 250–55.
- Stover, R. H., and N. W. Simmonds. 1987. Bananas. 3rd ed. London: Longman.
- Swait, J. R. 1994. A structural equation model of latent segmentation and product choice for cross-sectional revealed preference choice data. *Journal of Retailing and Consumer Services* 1: 77–89.
- Swennen, R., and D. R. Vuylsteke. 1988. Bananas in Africa: Uses and prospects for improvement. In *Crop genetic resources for Africa*. Vol. 2. Proceedings of an International Conference, Ibadan, Nigeria.
- Traynor, P. 2003. Uganda biotechnology project. A final draft report submitted to the U.S. Agency for International Development, RAISE Task Order no. 817.
- Tripathi, L. 2003. Genetic engineering for improvement of *Musa* production in Africa. *African Journal of Biotechnology* 2 (12): 503–08.
- Tushemereirwe, W. K., A. Kangire, J. Smith, F. Ssekiwoko, M. Nakyanzi, D. Kataama, C. Musiitwa, and R. Karyeija. 2003. An outbreak of banana bacterial wilt on banana in Uganda. *InfoMusa* 12 (2): 6–8.
- Tushemereirwe, W., I. K. Kashaija, W. Tinzaara, C. Nankinga, and S. S. New. 2000. *Banana production manual: A guide to successful banana production*. National Agricultural Research Organization and Agribusiness Development Centre Investment in Developing Export Agriculture (ADC-IDEA).
- Tushemereirwe, W. K., I. Kashaija, W. Tinzaara, C. Nankinga, and S. New. 2003. *Banana production manual: A guide to successful banana production in Uganda*. 2nd ed. Kampala: Makerere University Printery.
- Uganda Bureau of Statistics (UBOS). 2006a. 2002 Uganda population and housing census: Main report. http://www.ubos.org/2002%20Census%20Final%20Reportdoc.pdf.
- ———. 2006b. National Statistics Databank, Agriculture and Fisheries. Area planted and production of selected food crops, 1980–2004. http://www.ubos.org.
- Ugandan National Council for Science and Technology (UNCST). 2006. A gap analysis study of the communication and outreach strategy for biotechnology and biosafety in Uganda. Unpublished report.
- United Nations Environment Programme (UNEP). 2001. *International technical guidelines for safety in biotechnology*. http://www.biosafetyprotocol.be/UNEPGuid/Contents.html (accessed August 2007).

- Wafula, D., and N. Clark. 2005. Science and governance of modern biotechnology in Sub-Saharan Africa—the case of Uganda. *Journal of International Development* 17: 679–94.
- Wesseler, J. 2003. Resistance economics of transgenic crops: A real option approach. In *Battling resistance to antibiotics and pesticides*, edited by R. Laxminarayan, pp. 214–37. Washington, D.C.: Resources for the Future
- Wesseler, J., S. Scatasta, and E. Nillesen. 2007. The maximum incremental social tolerable irreversible costs (MISTICs) and other benefits and costs of introducing transgenic maize in the EU-15. *Pedobiologia* 51 (3): 261–69.

#### RECENT IFPRI DISCUSSION PAPERS

For earlier discussion papers, please go to www.ifpri.org/pubs/pubs.htm#dp.

All discussion papers can be downloaded free of charge.

- 766. Global macroeconomic developments and poverty. Eugenio Diaz-Bonilla, 2008.
- 765. Economic partnership agreements between the European Union and African, Caribbean, and Pacific countries: what is at stake for Senegal. Valdete Berisha-Krasniqi, Antoine Bouet and Simon Mevel, 2008.
- 764. Marriage behavior response to prime-age adult mortality: Evidence from Malawi. Mika Ueyama and Futoshi Yamauchi, 2008.
- 763. Impacts of inventory credit, input supply shops, and fertilizer microdosing in the drylands of Niger. John Pender, Tahirou Abdoulaye, Jupiter Ndjeunga, Bruno Gerard, and Edward Kato, 2008.
- 762. Plant genetic resources for agriculture, plant breeding, and biotechnology: Experiences from Cameroon, Kenya, the Philippines, and Venezuela. José Falck-Zepeda, Patricia Zambrano, Joel I. Cohen, Orangel Borges, Elcio P. Guimarães, Desiree Hautea, Joseph Kengue, and Josephine Songa, 2008.
- 761. How change agents and social capital influence the adoption of innovations among small farmers: Evidence from social networks in rural Bolivia. Mario Monge, Frank Hartwich, and Daniel Halgin, 2008.
- 760. Gender Difference in the Long-Term Impact of Famine. Ren Mu and Xiaobo Zhang, 2008.
- 759. Rural innovation systems and networks: Findings from a study of Ethiopian smallholders. David J. Spielman, Kristin E. Davis, Martha Negash, and Gezahegn Ayele, 2008.
- 758. A Tale of two countries: spatial and temporal patterns of rice productivity in China and Brazil. Liangzhi You, 2008.
- 757. Must conditional cash transfer programs be conditioned to be effective? The impact of conditioning transfers on school enrollment in Mexico. Alan de Brauw and John Hoddinott, 2008.
- 756. From the ground up: Impacts of a pro-poor community-driven development project in Nigeria. Ephraim Nkonya, Dayo Phillip, Tewodaj Mogues, John Pender, Muhammed Kuta Yahaya, Gbenga Adebowale, Tunji Arokoyo, and Edward Kato, 2008.
- 755. Child labor and schooling responses to production and health shocks in Northern Mali. Andrew Dillon, 2008.
- 754. Decentralization, local government elections and voter turnout in Pakistan. Kamiljon T. Akramov, Sarfraz Qureshi, Regina Birner, and Bilal Hasan Khan, 2008.
- 753. Regional biotechnology regulations: Design options and implications for good governance. Regina Birner and Nicholas Linacre, 2008.
- 752. The rise of supermarkets and their development implications: International experience relevant for India. Thomas Reardon and Ashok Gulati, 2008.
- 751. *Investing in African agriculture to halve poverty by 2015*. Shenggen Fan, Michael Johnson, Anuja Saurkar, and Tsitsi Makombe, 2008.
- 750. Accelerating growth and structural transformation: Ghana's options for reaching middle-income country status. Clemens Breisinger, Xinshen Diao, James Thurlow, Bingxin Yu, and Shashidhara Kolavalli, 2008.
- Seed value chains for millet and sorghum in Mali: A state-based system in transition. Lamissa Diakité, Amadou Sidibé, Melinda Smale, and Mikkel Grum, 2008.
- 748. *The impact of CAFTA on employment, production and poverty in Honduras*, Samuel Morley, Eduardo Nakasone, and Valeria Piñeiro, 2008.
- 747. Asian-driven resource booms in Africa: Rethinking the impacts on development. Clemens Breisinger and James Thurlow, 2008
- 746. Trading millet and sorghum genetic resources: Women vendors in the village fairs of San and Douentza, Mali. Melinda Smale, Lamissa Diakité Brahima Dembélé, Issa Seni Traoré, Oumar Guindo, Bouréma Konta, 2008.
- 745. Determinants of smallholder commercialization of food crops: Theory and evidence from Ethiopia. John Pender and Dawit Alemu, 2007.

# INTERNATIONAL FOOD POLICY RESEARCH INSTITUTE

# www.ifpri.org

# IFPRI HEADQUARTERS

2033 K Street, NW Washington, DC 20006-1002 USA

Tel.: +1-202-862-5600 Fax: +1-202-467-4439 Email: ifpri@cgiar.org

#### IFPRI ADDIS ABABA

P. O. Box 5689 Addis Ababa, Ethiopia

Tel.: +251 11 6463215 Fax: +251 11 6462927

Email: ifpri-addisababa@cgiar.org

#### **IFPRI NEW DELHI**

CG Block, NASC Complex, PUSA New Delhi 110-012 India

Tel.: 91 11 2584-6565

Fax: 91 11 2584-8008 / 2584-6572 Email: ifpri-newdelhi@cgiar.org